Author's response to reviews

Title: Traumatic spondyloptosis of the lumbar spine: A case report

Authors:

Robert Amesiya (drrobertamesiya@yahoo.com)
Mallon Nyati (mallonnyati@yahoo.com)
Norbert T Orwotho (otnorbert@yahoo.com)
Rodney Mugarura (rmugarura@gmail.com)
Erisa S Mwaka (erisamwaka@yahoo.com)

Version: 5
Date: 25 November 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Traumatic spondyloptosis of the lumbar spine: A case report

Authors:

Robert Amesiya:  drrobertamesiya@yahoo.com
Mallon Nyati:  mallonnyati@yahoo.com
Norbert Orwotho:  otnorbert@yahoo.com
Rodney Mugarura:  rmugarura@gmail.com
*Erisa S. Mwaka:  erisamwaka@yahoo.com

Version 2: 14 October 2014

Authors’ response to reviews: see over
The Biomed Central Editorial Team

**Object: MS: 396967138223120** Traumatic spondyloptosis of the lumbar spine: A case report. Dr Amesiya et al.

We would like to thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your reviewers’ comments.

**Editor’s comment**

We have changed ‘Manson’ to ‘Mason in the case report section of the manuscript